Author’s response to reviews

Title: Associations between male infertility and ancestry in South Americans: A case control study

Authors:

Maria Skowronek (fernandaskowronek@gmail.com)
Tatiana Velazquez (tavalentina@hotmail.com)
Patricia Mut (mut.patricia@gmail.com)
Gonzalo Figueiro (vazfigure@gmail.com)
Monica Sans (mbsans@gmail.com)
Bernardo Bertoni (bbertoni@fmed.edu.uy)
Rossana Sapiro (rsapiro@fmed.edu.uy)

Version: 3 Date: 10 Jul 2017

Author’s response to reviews:

Dr. Matteo Pasini
Editor in Chief
Anne Menard
BMC Medical Genetics

Dear Drs,

Please find our corrections of the manuscript entitled “Associations between male infertility and ancestry in South Americans: A case control study” by Skowronek et al. to be considered for publication as a research paper in BMC Medical Genetics.

We have introduced the indicated changes as follow:

1. We uploaded supplementary table as a supplementary files.

2. The contributions of TV, PM, GF and MS were deeply explained
3. Regarding quality of the English: we decided to use a professional language editing service. We asked American Journal Experts to review the manuscript. Notice that after English correction the title of the manuscript is: “Associations between male infertility and ancestry in South Americans: A case control study”

4. We uploaded our manuscript, tables, figures and additional files as final, clean version without any track.

We thank you for your consideration,

Sincerely,

Dr. Sapiro

Rossana Sapiro M.D. PhD
Histology and Embriology Department
School of Medicine,
Avenida General Flores 2125, CP 11800,
Montevideo-Uruguay,
Tel: (598-2)-9243414/3551,
Fax: (598-2)-9249563,
e-mail: rsapiro@fmed.edu.uy